Author's response to reviews

Title: A randomised control trial of low glycaemic index carbohydrate diet vs. no dietary intervention in the prevention of recurrence of fetal macrosomia.

Authors:

Jennifer M Walsh (msjenniferwalsh@gmail.com)
Rhona Mahony (rhonamahony@hotmail.com)
Michael Foley (mfoley@nmh.ie)
Fionnuala McAuliffe (fionnuala.mcauliffe@ucd.ie)

Version: 2 Date: 15 January 2010

Author's response to reviews: see over
Dear Sir/Madam,

I have attached a revised manuscript outlining a proposed randomised control trial.

The following points have been revised as suggested.

1. The word ‘Title’ has been removed from the title
2. A Title page has been included
3. The trial registration number has been added to the foot of the abstract
4. A ‘Competing interests’ section has been included
5. An ‘Author’s contributions’ section has been added

We would like to thank you for your ongoing review of this manuscript and will be happy to deal with any further queries.

Yours Sincerely,

Jennifer Walsh

Dr Jennifer Walsh

Clinical Research Fellow
University College Dublin
National Maternity Hospital
Holles Street
Dublin 2
Ireland

Professor Fionnuala McAuliffe

UCD Obstetrics & Gynaecology,
UCD School of Medicine and Medical Science,
University College Dublin,
National Maternity Hospital,
Holles St,
Dublin 2
Ireland